Author's response to reviews

Title: Early Motherhood: A Qualitative Study Exploring the Experiences of African Australian Teenage Mothers in Greater Melbourne, Australia *1 Mimmie Claudine Ngum Chi Watts, MPH, PhD Lecturer, College of Health and Biomedicine, Victoria University, Melbourne, Australia P O Box 14428 Melbourne, Australia. Tel: +613 9919 2766. Mimmie.ngumchi@vu.edu.au 2 Pranee Liamputtong, PhD Professor, School of Public Health & Human Biosciences, Faculty of Health Sciences, La Trobe University, Bundoora, Victoria, Australia 3086. Pranee@latrobe.edu.au 3 Celia McMichael, PhD Lecturer, School of Social Sciences and Communications, La Trobe University, Bundoora, Victoria, Australia 3086 c.mcmichael@latrobe.edu.au *1 Corresponding Author

Authors:

Mimmie C Ngum Chi Watts Dr (mimmie.ngumchi@vu.edu.au)
Pranee Liamputtong Prof (Pranee@latrobe.edu.au)
Celia McMichael Dr (C.mcmichael@latrobe.edu.au)

Version: 4 Date: 28 January 2015

Author's response to reviews:

Dear Editor,

I have revised the manuscript as advised. I have included the title page with the abstract in the manuscript

Sincerely,
Dr Mimmie Watts